Author's response to reviews

Title: Low penetrance of retinoblastoma for p.V654L mutation of the RB1 gene

Authors:

Chia-Cheng Hung (chiacheng@ntu.edu.tw)
Shin-Yu Lin (lin.shinyu@gmail.com)
Chien-Nan Lee (leecn@ntu.edu.tw)
Chih-Ping Chen (cpc_mmh@yahoo.com)
Shuan-Pei Lin (zsplin14@gmail.com)
Mei-Chyn Chao (mcchao@kmu.edu.tw)
Shyh-Shin Chiou (chiouss@kmu.edu.tw)
Yi-Ning Su (ynsu@ntu.edu.tw)

Version: 3 Date: 25 May 2011

Author's response to reviews:

We revised our manuscript according to the review's suggestion.
1) In fig. 1 individual III3 is not marked * although according to the text he tested positive for the mutation.
2) the correct nomenclature for the mutation is p.V654L.